Author's response to reviews

Title: Ofloxacin Plus Rifampicin Versus Doxycycline Plus Rifampicin In The Treatment Of Brucellosis. A Randomized Clinical Trial

Authors:
Dr Oguz Karabay (drkarabay@yahoo.com)
Dr Irfan Sencan (isencan@ibuduzce-tip.edu.tr)
Derya Kayas (deryakayas@yahoo.com)
Idris Sahin (isahin@ibuduzce-tip.edu.tr)

Version: 3 Date: 3 Nov 2003

"BMC Infectious Diseases"

Dear Editor,

I stated that the current instructions to authors have been read, thereby indicating compliance with those instructions and acceptance of the conditions posed.

All the named authors have seen and agreed to the submitted version of the paper "OFLOXACIN PLUS RIFAMPICIN VERSUS DOXYCYCLINE PLUS RIFAMPICIN IN THE TREATMENT OF BRUCELLOSIS. A RANDOMIZED CLINICAL TRIAL" that who have been acknowledged as contributors or as providers of personal communications have agreed to their inclusion, that the material is original and it has never been neither published elsewhere nor submitted for publication simultaneously.

If it is accepted for publication, the authors transfer copyright of manuscript to the "BMC Infectious Diseases".

Sincerely

ASSIST PROF, OGUZ KARABAY, MD
ASSIST PROF, IRFAN SENCAN, MD
ASSIST PROF, IDRIS SAHIN, MD
Dr. Derya Kayas, MD